Author's response to reviews

Title: Anaphylactic reaction associated with Ranitidine in a patient with acute pancreatitis: A Case report.

Authors:

Ulfirn S Rethnam (ulfinr@yahoo.com)
Rajam S Yesupalan (ajeesh2000@yahoo.co.uk)

Version: 4 Date: 5 August 2007

Author's response to reviews:

To
The Editor,
Journal of Medical Case Reports

SUB: Revised Case report for publication in your esteemed journal

Sir,

We would like to submit the revised case report titled "Anaphylactic reaction associated with Ranitidine in a patient with acute pancreatitis: A case report" for publication in your esteemed journal. Following is a point to point response to the concerns raised by the reviewers.

Reviewer 1:
1. The comment as suggested by reviewer 1 has been added to the manuscript in the discussion section.

Reviewer 2:
1. Incidence of ranitidine anaphylaxis as reported in the literature and the relevant reference has been added to the manuscript.
2. The term "significant family history" has been changed to "family history of drug allergies" which is self explanatory.
3. The consent has been obtained from the patient as the patient survived the anaphylactic reaction.
4. The hospital pharmacy and the dermatology departments were informed of this adverse reaction who would have informed the national authorities.

Please do consider this report for publication.

Thanking you,

Yours truly,

Mr Ulfin Rethnam